A plain language summary on assessing the long-term effectiveness of cladribine tablets in people living with relapsing multiple sclerosis: The CLASSIC-MS study

Gavin Giovannoni¹, Alexey Boyko², Jorge Correale³, Gilles Edan⁴, Mark S Freedman⁵, Xavier Montalban⁶, Kottil Rammohan⁷, Dusan Stefoski⁸, Bassem Yamout^{9,10}, Thomas Leist¹¹, Aida Aydemir¹², Laszlo Borsi¹³, Elisabetta Verdun di Cantogno¹²

¹Blizard Institute, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK; ²Pirogov Russian National Research Medical University, Department of Neurology, Neurosurgery and Medical Genetics, Federal Center of Brain Research and Neurotechnologies, Moscow, Russia; ³Department of Neurology, FLENI Institute, Buenos Aires, Argentina; ⁴Department of Neurology, University Hospital of Rennes, Rennes, France; ⁵University of Ottawa, Department of Medicine and the Ottawa Hospital Research Institute, Ottawa, ON, Canada; ⁶Department of Neurology-Neuroimmunology, Centre d'Esclerosi Múltiple de Catalunya (Cerncat), Hospital Universitario Vall d'Hebron, Barcelona, Spain; ⁷University of Miami School of Medicine, MS Research Center, Miami, FL, USA; ⁸Department of Neurological Sciences, Rush Medical College, Chicago, IL, USA; ⁹Neurology Institute, Harley Street Medical Center, Abu Dhabi, UAE; ¹⁰American University of Beirut Medical Center, Beirut, Lebanon; ¹¹Division of Clinical Neuroimmunology, Jefferson University, Comprehensive MS Center, Philadelphia, PA, USA; ¹²EMD Serono Research & Development Institute, Inc., Billerica, MA, USA (an affiliate of Merck KGaA); ¹³Merck Healthcare KGaA, Darmstadt, Germany

First draft submitted: 12 May 2023; Accepted for publication: 14 July 2023; Published online: 3 August 2023

Summary

What is this summary about?

Previous studies have shown that people living with multiple sclerosis (MS) treated with cladribine tablets have fewer relapses (where new symptoms occur or existing symptoms get worse for 24 hours or more) and delayed disability progression (slowing down of the disease getting worse).

The CLASSIC-MS study looked at the long-term effectiveness of treatment with cladribine tablets in people living with MS who had taken part in the original CLARITY and CLARITY Extension clinical studies.

What were the results?

Results showed that people treated with cladribine tablets maintained their mobility (the ability to move freely) for longer and experienced other positive effects long after their treatment ended, including being less likely to need further treatment for their MS.

What do the results mean?

The results obtained from CLASSIC-MS show that the benefits of taking cladribine tablets carry on even when patients stop taking the treatment.

Where can I find the original article on which this summary is based?

The original article is called 'Long-term follow-up of patients with relapsing multiple sclerosis from the CLARITY/CLARITY Extension cohort of CLASSIC-MS: An ambispective study'. You can read the article for free here: <u>https://journals.sagepub.com/doi/full/1</u> 0.1177/13524585231161494?rfr dat=cr pub++0pubmed&url ver=Z39.88-2003&rfr id=ori%3Arid%3Acrossref.org

How to say (double click sound icon to play sound)...

- Cladribine: CLAD-ree-BEEN
- Multiple sclerosis:
- MUHL-tuh-pl-sklr-OW-suhs
- Myelin: Mai-UH-lin
- Placebo: pluh-SEE-boh
- Relapse: RE-lapse

ient endea, including being less like

Future Medicine part

Neurodegenerative Disease Management



Who sponsored this study?

Financial support for this summary was provided by Merck (CrossRef Funder ID: 10.13039/100009945).

Who should read this article?

The authors of the original publication developed this summary to help people living with multiple sclerosis, their caregivers, healthcare professionals, and patient advocates to understand the results of this study.

The original studies

CLARITY: The study looked at whether cladribine tablets work well for people with relapsing-remitting multiple sclerosis, including its side effects and tolerability (the degree to which a drugs' side effects can be put up with by a patient). It was carried out between April 2005 and November 2008.

CLARITY Extension: This extension study further tested the side effects and tolerability of cladribine tablets in people who have previously completed treatment within CLARITY. It was carried out between February 2008 and December 2011.

What is multiple sclerosis?

Multiple sclerosis (also known as MS) is a potentially disabling disease where the body's own immune cells (the body's own defence cells) attack the **central nervous system** (CNS). The reasons why this happens are not yet understood.

Central nervous system: The central nervous system (CNS) is made up of the brain and spinal cord. The CNS controls most functions of the body through nerves.

Cognition: Cognition refers to 'thinking' processes such as attention, learning, memory, understanding, planning, and making decisions. About half of people with MS experience changes in their cognitive abilities. Sometimes people with MS describe the experience as 'brain fog', when they may be struggling to organise their thoughts.

Immune cells damage the CNS by attacking the nerves and the fatty covering (known as myelin) that protects the nerves. This means it is more difficult for nerve signals to travel through the body in someone with MS, resulting in a range of possible symptoms.

People living with MS may experience problems with movement, fatigue (lack of energy), eyesight, or **cognition** (thinking, learning, and planning). These are just some of the examples of the possible symptoms of MS.



There is currently no cure for MS. However, disease-modifying therapies (treatments that can change how your MS develops over time) have been proven to help ease symptoms and delay the course of the disease.

There are different types of MS. The most common type is called relapsing-remitting MS. It affects around 85% (almost 9 out of 10) of people first diagnosed with the illness. People living with relapsing-remitting MS have relapses (where new symptoms occur or existing symptoms get worse for 24 hours or more), followed by a period of full or partial recovery (feeling better).

What are cladribine tablets?

Cladribine tablets (Mavenclad[®]) are an oral medicine (given by mouth) for treating relapsing types of MS.

The treatment works by reducing the numbers of certain immune cells, called B and T **lymphocytes** (white blood cells).

These lymphocytes are thought to be the main immune cells involved in MS, and so MS treatments often target them.

Lymphocytes: Lymphocytes are types of white blood cells that are part of the immune system. Lymphocytes include B cells, T cells, and natural killer cells. B and T cells work together to fight infections.

Cladribine tablets are given for a short amount of time followed by a long break. To explain in more detail, patients receive

8–10 days of treatment, then they have a break of 12 months. After the 12 months have passed, patients receive another treatment course of 8–10 days.

Studies on cladribine tablets have shown that the treatment works to lower the number of relapses in people living with MS and slows the worsening of their illness.



Cladribine tablets are administered as two courses, based on the patient's weight, separated by 1 year

Maximum of 20 days of treatment

Why was the CLASSIC-MS study carried out?

The original CLARITY and CLARITY Extension clinical studies showed that cladribine tablets lowered the number of relapses that people living with MS have. The studies also showed that treatment helped to lower the risk of people's disabilities getting worse.

People took part in CLARITY and CLARITY Extension several years ago. The same people then took part in the CLASSIC-MS study, and so researchers had the chance to find out how well treatment with cladribine tablets had worked after the original 2-year treatment course.

• On average, people took part in the CLASSIC-MS study around 11 years after their last dose of treatment received during the CLARITY or CLARITY Extension studies

How was the CLASSIC-MS study carried out and who took part?

A total of 435 people were included in CLASSIC-MS from the original CLARITY and CLARITY Extension studies.

- They were an average age of 53 years and were mainly women (68% or almost 7 out of 10)
- 91% (just over 9 out of 10) had received cladribine tablets in CLARITY or CLARITY Extension, with 160 people having
 received the prescribed dose of 3.5 mg/kg over 2 years. The other 234 people received different doses of cladribine tablets

Expanded Disability Status Score (or EDSS): This is a way of measuring disability in people living with MS that considers aspects such as vision, coordination, and mobility. The score is widely used in clinical studies and ranges from 0 to 10; a higher score represents a higher level of disability.

The score is often summarised as the median, which is the middle number when the values are sorted from smallest to largest.

during the studies

Concerning their level of disability, people had a median **Expanded Disability Status Score** (EDSS) of 3.5 at the start of CLASSIC-MS; it was 2.5 at the start of either CLARITY or CLARITY Extension.

What did the researchers measure in CLASSIC-MS?

Long-term mobility

• This was defined as no wheelchair use in the 3 months before a person's first visit in CLASSIC-MS, and not bedridden at any time since their last dose of treatment in CLARITY or CLARITY Extension (meaning that the person had an EDSS less than 7)

Ż

Long-term disability status

• This concerned people with no use of a walking aid (such as a cane or frame) at any time since their last dose of treatment in CLARITY or CLARITY Extension (meaning the person had an EDSS less than 6)

Real-world treatment patterns and outcomes, since taking part in CLARITY or CLARITY Extension

- · Whether people used any further disease-modifying medication for their MS
- The number of relapses that people experienced
- The employment status of people living with MS who enrolled in CLASSIC-MS

What were the results from the study?

Long term mobility

- Most people were not using a wheelchair in the 3 months before they took part in CLASSIC-MS and were not bedridden at any time since their last dose of treatment received during the original studies. This included 90% (9 out of 10) of the people who took cladribine tablets in CLARITY or CLARITY Extension and 78% (almost 8 out of 10) of people who did not
- Considering the people who took the prescribed dose of cladribine tablets (3.5 mg/kg over 2 years), this figure was 88% (almost 9 out of 10)



Long-term disability status

- Most people taking part in CLASSIC-MS did not use a walking aid at any time since their last dose of treatment received during the original studies. This included 81% (just over 8 out of 10) of the people who took cladribine tablets in CLARITY or CLARITY Extension and 76% (almost 8 out of 10) of those who did not
- Considering the people who took the prescribed dose of cladribine tablets (3.5 mg/kg over 2 years), this figure was 79% (almost 8 out of 10)



Treatment patterns and outcomes

- Around half of people taking part in CLASSIC-MS did not use any further treatment for their MS since their last dose of treatment received during the original studies. This included 56% (almost 6 out of 10) of the people who took cladribine tablets in CLARITY or CLARITY Extension and 27% (almost 3 out of 10) of those who did not
 - For people who received the prescribed dose of cladribine tablets (3.5 mg/kg over 2 years), 58% (almost 6 out of 10) did not use any further treatment for their MS



- In the total study population, 46% (almost 5 out of 10) were relapse-free at the time of the CLASSIC-MS study. This included 48% (almost 5 out of 10) of the people who took cladribine tablets in CLARITY or CLARITY Extension and 27% (almost 3 out of 10) of those who did not
 - For people who received the prescribed dose of cladribine tablets (3.5 mg/kg over 2 years), 47% (almost 5 out of 10) were relapse-free



- Finally, the investigators asked people if they were employed in some way at the start of CLASSIC-MS. This included 51% (just over 5 out of 10) of the people who took cladribine tablets in CLARITY or CLARITY Extension and 28% (almost 3 out of 10) of those who did not
 - In the group of people who received the prescribed dose of cladribine tablets (3.5 mg/kg over 2 years), 54% (just over 5 out of 10) were employed in some capacity



What do the results of this study mean?

The findings show that people treated with cladribine tablets had a lower risk of their disability getting worse compared with people who did not take the treatment. People taking cladribine tablets were also less likely to need further treatment for their MS.

The results from CLASSIC-MS show that the benefits of cladribine tablets carry on even when patients stop taking the treatment.

Where can readers find more information on this study?

This is a summary of an article called "Long-term follow-up of patients with relapsing multiple sclerosis from the CLARITY/ CLARITY Extension cohort of CLASSIC-MS: An ambispective study", originally published in *Multiple Sclerosis Journal*. You can read the full article here: <u>https://journals.sagepub.com/doi/10.1177/13524585231161494</u>

The original publication citation is: Giovannoni G, Boyko A, Correale J *et al*. Long-term follow-up of patients with relapsing multiple sclerosis from the CLARITY/CLARITY Extension cohort of CLASSIC-MS: An ambispective study. *Multiple Sclerosis Journal* 2023; Epub ahead of print. doi: 10.1177/13524585231161494. <u>ClinicalTrials.gov</u> identifier: NCT03961204

Educational resources

Read more about multiple sclerosis, including symptoms and criteria for diagnosis, treatment options, and general support for those living with multiple sclerosis, at these websites:

- National Multiple Sclerosis Society available at: <u>https://www.nationalmssociety.org</u>
- Multiple Sclerosis Trust available at: https://mstrust.org.uk/

Plain Language Summary of Publication Giovannoni, Boyko, Correale and co-authors

Financial & competing interests disclosure

Full author disclosure and ethical information can be found in the original article. The authors had full control of this summary and provided their final approval of all content.

Plain language summary writing and editorial support was provided by Joseph Ward of inScience Communications, Springer Healthcare Ltd., UK, and was funded by Merck.